Author’s response to reviews

Title: A nurse-delivered mental health intervention for obstetric fistula patients in Tanzania: Results of a pilot randomized controlled trial

Authors:

Melissa Watt (melissa.watt@duke.edu)
Mary Mosha (maryanfort@yahoo.com)
Alyssa Platt (alyssa.platt@duke.edu)
Kathleen Sikkema (kathleen.sikkema@duke.edu)
Sarah Wilson (sarah.wilson@duke.edu)
Elizabeth Turner (elizabeth.turner@duke.edu)
Gileard Masenga (drgmasenga@yahoo.com)

Version: 2 Date: 07 Aug 2017

Author’s response to reviews:

Thank you for reviewing our revised manuscript, and for picking up on these issues requiring changes. I have now addressed the two reviewer comments:

1. Page 9, lines 1-2: one of the questions regarding intervention satisfaction is repeated, likely excluding one of the five questions indicated.

The second repeated question was meant to be "satisfaction of intervention facilitator". This has now been revised.

2. Page 9, lines 9-10: per previous round of reviewer comments this should read symptomatology as has been done under the next subheading 'clinical significance analysis'.

This has been revised to incorporate the term symptomatology.